Author’s response to reviews

Title: Clinical comparative effectiveness of acupuncture versus manual therapy treatment of lateral epicondylitis: feasibility randomized clinical trial.

Authors:
Katrine Bostrøm (katrine.bostrom@gmail.com)
Sverre Mælhum (sverre.maehlum@nimi.no)
Milada Cvancarova Småstuen (milasm@oslomet.no)
Kjersti Storeheim (kjersti.storheim@medisin.no)

Version: 1 Date: 05 Mar 2019

Author’s response to reviews:

Dear Editor,

Thank you for inviting us to revise and resubmit the manuscript: "Clinical comparative effectiveness of acupuncture versus manual therapy treatment of lateral epicondylitis: a randomized pilot study for a three-armed controlled trial." (PAFS-D-18-00155).

We thank the Deputy Editor and both reviewers for their thorough evaluation and the thoughtful comments on the manuscript. Below is a point-by-point response to all comments. We have responded to each comment made by the reviewers successively in this letter, each comment is cited and is followed by our response and actions.

We hope that you consider our manuscript much improved and suitable for publication in your journal.